





Citation: Arata L., Cerroni S., Santeramo F.G., Trestini S., Severini S. (2023). Towards a holistic approach to sustainable risk management in agriculture in the EU: a literature review. *Bio-based and Applied Economics* 12(3): 165-182. doi: 10.36253/bae-14492

Received: March, 4 2023 Accepted: September, 11 2023 Published: October 15, 2023

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Competing Interests: The Author(s) declare(s) no conflict of interest.

Editor: Donato Romano Discussant: Roberta Raffaeli, Paolo Sckokai

ORCID

LA: 0000-0002-3011-5381 SC: 0000-0001-9231-1591 FGS: 0000-0002-9450-4618 ST: 0000-0002-9828-8262 SS: 0000-0001-5501-3552 BAE 10th Anniversary papers

Towards a holistic approach to sustainable risk management in agriculture in the EU: a literature review

Linda Arata¹, Simone Cerroni², Fabio Gaetano Santeramo³, Samuele Trestini⁴, Simone Severini^{5,*}

- ¹ Università Cattolica Del Sacro Cuore, Piacenza, Italy
- ² University of Trento, Trento, Italy
- ³ University of Foggia, Foggia, Italy
- ⁴ University of Padova, Padova, Italy
- ⁵ University of Tuscia, Viterbo, Italy
- *Corresponding author. E-mail: severini@unitus.it

Abstract. Agriculture is one of the sectors most exposed to a plethora of risky phenomena such as weather, pests and diseases, changes in prices and government policies, instability of global markets. We review the literature on risk management (RM) in agriculture focusing on five key issues: i) why evidence on RM is often controversial; ii) how farmers behave in selecting among available RM instruments; iii) why some of these instruments are underutilised; iv) how to assess the impacts of innovative RM tools to (further) improve their design; v) how agricultural policy measures aimed at increasing the environmental sustainability of the sector could affect RM choices. We address all these issues to get a holistic vision of RM, and point at areas where further analyses are needed.

Keywords: risk management choices, behavioural factors, adoption of risk management tools, use of chemicals, feasibility studies.

JEL Code: Q12, D81, D83, O31.

1. INTRODUCTION

Although risk concerns all economic activities, agriculture is one of the most concerned sectors, due to its exposure to a plethora of risky phenomena such as weather, pests and diseases, changes in prices and government policies, instability of global markets, and other factors (Moschini and Hennessy, 2001; Hardaker et al., 2015; Komarek et al., 2020). Furthermore, the multifaceted risks farmers must cope with are very likely to occur simultaneously, producing a compounded negative effect (Hardaker et al., 2004).

Risk in agriculture causes wide volatility in farmers' income and wellbeing and in turn it influences the decision-making process. Experiencing negative events reduces farmers' willingness to invest and innovate (Sckokai and Moro, 2009). This, in turn, may negatively affect farms' productivity and

Bio-based and Applied Economics 12(3): 165-182, 2023 | e-ISSN 2280-6172 | DOI: 10.36253/bae-14492

Copyright: © 2023 Arata L., Cerroni S., Santeramo F.G., Trestini S., Severini S.

Open access, article published by Firenze University Press under CC-BY-4.0 License.

competitiveness (Vigani and Kathage, 2019) and push farms out of the business. The negative consequences may also be reflected in the value chain (Cafiero, 2008) and transferred to all stakeholders of the agro-food system. Major and unexpected events such as the COV-ID-19 pandemic and the food/energy crises induced by the war in Ukraine have unrevealed the vulnerability of the global food supply. By threatening the status of global food security, these major shocks have induced unprecedented policy responses in all advanced economies, as well as in developing countries (European Parliament, 2022; OECD, 2020; Santeramo and Kang, 2022). Over the years, risk in agriculture has been increasing in width and depth, unveiling the need for improving Risk Management (RM), as recognized by the European Commission (2017) "[...] it is important to set up a robust framework for the farming sector to successfully prevent or deal with risks and crises, with the objective of enhancing its resilience and, at the same time, providing the right incentives to crowd-in private initiatives". RM refers to the actions taken to manage potential problems induced by risky events, to reduce their detrimental consequences, and to increase the chances of success of the business (Kahan, 2013). In this sense, RM can be a key factor in enhancing the resilience of farms and related farming systems (Spiegel et al., 2020) and several scholars call for improving and enlarging the scope of RM to do so (Finger et al., 2022). Unfortunately, the state of knowledge on RM in agriculture is still incomplete, and the current approaches to RM are too simple, partial, and inappropriate to successfully help cope with multi-faced global challenges: changes in climate, more frequent extreme weather events, unstable and volatile markets, food security and food safety threats. Improving the state of knowledge on RM is important: successfully managing risks helps in finding the right balance among productivity, environmental care, market resilience to climate change, and capability to secure safe and quality food.

This paper reviews the extant literature on the analyses of agricultural RM, highlights progress and gaps, and advices on promising areas of research. This exercise is *per se* a very useful contribution to developing a holistic approach to analysing RM. More generally, we hope this piece will stimulate the debate on this relevant topic. While we are aware that some recent literature reviews exist, especially on specific topics (e.g., Komarek et al., 2020), we believe that our paper makes a twofold contribution to the extant debate. First, our overview of the literature focuses on five research questions: i) why evidence on RM is often controversial; ii) how farmers behave in selecting among available RM instruments; iii) why some of these instruments are underutilised; iv) how to assess the impacts of innovative RM tools to (further) improve their design; v) how agricultural policy measures aimed at increasing the environmental sustainability of the sector could affect risk and, consequently, RM choices. These questions are answered in the subsequent sections. This review also highlights areas where further analyses are needed. Second, we use a holistic approach to the topic. Since RM in agriculture is a complex phenomenon, several RM actions are available, and farmers' decisions are affected by spatially and temporally heterogeneous factors, a holistic approach seems needed (Figure 1). RM decisions are strongly influenced by the context in which farmers operate. Several dimensions are relevant to define the context, including not only farm structural and productive characteristics, but also the markets and the environment in which farmers operate. Regarding the markets, the complexity and interconnection of the global agri-food sector have imported new risks into the sector or emphasized old ones. Regarding the environment, there is a vast literature pointing out the effect of climate change on the risks farmers are facing (e.g., Sorvali et al., 2021). A growing body of literature has also shown that farmer's behavioural factors do affect the farmer's RM choices and therefore such factors cannot be ignored. Furthermore, the farm sector in the EU is heavily affected by policies. On the one hand, EU rural development policies support the adoption of specific RM tools providing subsidies to reduce the cost of adoption. On the other hand, farm production is constrained by pieces of legislation aimed at reducing the use of inputs with a harmful effect on the environment. However, often these inputs (e.g., pesticides in the case of pests, and irrigation in the case of drought) have also an effect on agricultural risk, thus their imposed reduction is likely to influence RM choices. The policy context is evolving in this area: the recently released Farm-to-Fork strategy (F2FS) and the CAP reform (European Commission, 2018) have set very ambitious environmental targets for EU agriculture (reduction of 50% and 20% in the use of pesticides and fertilisers respectively, by 2030). This will have consequences on the risk faced by farmers because the use of chemicals is intimately related to risk in agriculture and its management (Möhring et al., 2020). Studying the impact of policies targeted to environmental objectives on the farmer's risk and the uptake of RM tools is worthy to be addressed. Farmers are the ultimate decision-makers in terms of risk management strategies. As economic agents they can take several actions to manage risk including the adoption of specific RM tools (Santeramo, 2019; Cai, de Janvry and Sadoulet, 2020), changes in production mix and diversification, subscription of



Figure 1. Graphical representation of the environment in which RM takes place. Source: Own elaboration.

production contracts, use of risk decreasing input such as pest control chemicals and irrigation (Cerroni, 2020). Their actions are however influenced by risk preferences (Iyer et al., 2020), and other behavioural factors. The literature on the influence of other behavioural factors (i.e., subjective probabilities, risk perception and preferences, ambiguity attitudes, loss aversion and time preferences) on farmers' decisions to uptake RM tools (Colen et al., 2016) is scant (Coletta et al., 2018; Cerroni, 2020; Čop et al., 2023). Similar considerations apply to the attitude toward innovations and the ability to gather and process information. In the end, this oversimplified framework (and logical flow) advocates for a holistic approach to the analysis of RM also realizing that the current state of agricultural RM is constantly evolving, and it needs to be adapted to novel challenges. Our literature review is an attempt to approach the study of RM by adopting a holistic view: the methods adopted in the analysis of RM in agriculture, the behavioural factors affecting RM adoption, innovative RM tools, the relationship between agricultural risk and input use and between different policies directly or indirectly affecting risk and RM in agriculture. These topics were selected because they are important in influencing the choice of RM strategies. Furthermore, on these issues there have been significant advances in the literature but still some aspects require further study. Although the emphasis is on EU RM, the review considers analyses carried out also in non-EU countries. These are included to report approaches of analyses that could be replicated in the EU context and to better position the possible strategic choices of the EU with respect to the international context.

Section 2 gives an overview of the methods available in the literature to study the adoption of RM in agriculture and it also summarises risk types whose frequency is increasing in the agricultural sector. Section 3 focuses on the behavioural factors, specifically the subjective probabilities, risk and uncertainty preferences, affecting the farmer's decision to adopt insurance products (either traditional insurance or weather-based index insurance). Section 4 provides a picture of innovative RM tools, such as the mutual funds for catastrophic events (introduced by the last CAP reform) and the weatherbased index, together with their pros and cons compared to traditional insurance products. Section 5 reports on the studies related to the impact of pesticide and fertiliser use on agricultural risk as well as on the effect of insurance product adoption on this use. The section also highlights potential synergies and trade-off among different EU agricultural policies.

The last section concludes by summarizing the main points raised by the literature review. Here, special attention is paid to identify the areas where further improvements in the research on farm risk management are needed.

2. EXPLAINING RISK MANAGEMENT CHOICES

To cope with the large array of risks the farm sector is facing, the European Union (EU) decided to emphasize the role of new RM tools (Meuwissen et al., 2018) by structurally supporting not only crop insurance products, but also mutual funds (MF). These can cover yield losses and, by means of the Income Stabilisation Tool (IST), can help farmers cope with income drops (El Benni et al., 2016), enlarging the type of risks covered by subsidized tools. Despite the pervading exposure to risks for farmers (Trestini et al., 2017), the advantages that these instruments provide to farms (Enjolras et al., 2014; Severini et al., 2019) and the confirmed trend in the reduction of decoupled direct payments, the expenditure foreseen since 2014 in the CAP for the Risk Management Toolkit involves only 12 over 28 Member States (Chartier et al., 2017). Among these, Italy, the leading country in terms of allocated budget, still records a limited uptake of risk management tools (Ismea, 2022). The application of CAP is further pushing in the direction to improve the development and the support for risk management solutions by confirming actual tools and introducing in the Italian Strategic Plan of the CAP, from 2023, the new catastrophic risk coverage called Agricat, built as a mutual fund.

All this offers farmers the opportunity to get access to a wide set of RM solutions. This availability of innovative tools (i.e. mutual funds, IST and Agricat) together with the limited diffusion of traditional ones (i.e. insurance), impose understanding determinants of the diffusion of both traditional and innovative RM tools to allow farmers to maintain and improve their resilience and competitiveness under the new orientations of CAP. Understanding the factors that influence the adoption of RM tools allows, from the point of view of policy maker, to evaluate the effectiveness of RM policies and to guide their design. While, from the perspective of insurance company or mutual fund, it provides a better understanding of farm preferences by driving the development of RM tools that can promote farm resilience.

The review of the research methodologies applied to understanding the adoption of RM tools by farmers allows to identify research gaps and suggest potential future studies. The adoption of risk management tools is extensively investigated in the literature (Harrison and Ng, 2019), yet the behavioural factors of this adoption is often neglected (see Section 3 for a detailed discussion on this point). In the EU, a growing body of studies about yield insurance and mutual fund adoption is observable (Enjolras and Sentis, 2011; Liesivaara and Myyrä, 2017; Meuwissen et al., 2018; Santeramo, 2018; Santeramo et al., 2016; Was and Kobus, 2018). Indexbased insurance tools, marginally adopted in the EU, are mainly investigated in developing countries (e.g., Bucheli et al., 2021; Jensen et al., 2018) with some applications in the EU (Vroege et al., 2019).

In order to understand RM behaviour and assess the probability of farmers' adoption of available and innovative tools, it is worth considering different determinants simultaneously within an effective conceptual framework. In fact, the determinants of the adoption of RM tools are widely discussed in the literature, whereas practical application aiming to understand the interaction of different determinants is much less explored (Holt and Laury, 2002; Franken et al., 2017). Indeed, risk and ambiguity preferences may affect risk behaviour directly (Menapace et al., 2013; Čop et al., 2023). Furthermore, risk attitude explains risk behaviour, being indirectly affected by socio-economic and individual characteristics (Dohmen et al., 2011; Donkers et al., 2001). Complex interrelations can be simplified by multivariate statistical analysis such as the so-called Structural Equation Modelling (SEM) (Ullman and Bentler, 2003). SEM allows for testing complex models that imply both direct and indirect effects, allowing to solve limitations of traditional regression models. Furthermore, SEM allows researchers to distinguish between observed and latent variables, testing a wider variety of hypotheses compared with most traditional approaches (Kline, 1998). This approach is quite recent among agricultural economists, with one of the first examples incorporating risk components proposed by Pennings and Leuthold's (2000). In recent applications, risk behaviour has been investigated focusing on risk perception and risk attitude (Van Winsen et al., 2016), and on farm socio-economic and individual characteristics of risk attitude (Franken et al., 2017), also incorporating the role of trust and perceived barriers (Giampietri et al., 2020). A first attempt to apply a defined framework to understand the participation to RM tools has been applied by Rippo and Cerroni (2023) using the Unified Theory of Use and Acceptance (Venkatesh et al., 2003). Beside these attempts to build a framework able to understand and support the diffusion of available and innovative tools, any shared conceptual framework is, to our knowledge, applied in the literature.

Literature proposing the analysis of determinants of diffusion and/or adoption of innovation at SMEs including farms is extensive and well formalized. Many other methodologies should be tested also for the case of RM tools adoption. To do so, we should consider the organisational profile of farms and the role of individuals, especially in family farms. At first glance, when evaluating family farm choices, the theory of planned behaviour (TPB) (Ajzen, 1985; 1991) and technology acceptance model (TAM) (Davis, 1986) appear to be the most appropriate frameworks. This is because the two frameworks include constructs such as "Subjective norms", in the case of the TPB, or the perceived usefulness, in the case of the TAM, which strictly refer to the individual evaluation of the choice or to the perceived social pressure to engage or not to engage in a behaviour. When the farm adopts a structure in which wage labour becomes prevalent by assuming a corporate structure with division and delegation of responsibilities, the process of choice moves from being individual or dependent on family needs and relationships to being the result of an organizational choice. In this case, diffusion of innovation theory (DOI) (Rogers, 1995), and the technologyorganization-environment (TOE) framework (Tornatzky

and Fleischer, 1990) are suitable. These methodologies analyse the adoption process at the organizational level, including among determinants of specific adoption choice variables like the compatibility with the company, in the case of DOI, and formal/informal organisational link, in the case of TOE.

Besides deterministic approaches, Machine Learning solutions start to be applied to further understand factors affecting the adoption of RM strategies. Few applications can be retrieved from the current literature considering application of insurance contracts in Romania and Italy (Mare et al., 2022; Biagini et al., 2022a) and mutual funds for pest diseases in the North of Italy (Höschle et al., 2023).

At present, there is a growing interest in the management of new and/or growing risks: among these, it is worth to mention the need to manage growing systematic abiotic risks (e.g., drought and frost) and emerging biotic threats (pests and diseases). As regards the latter, the need for better-tailored risk management tools becomes more urgent given the orientation of agricultural policy towards the significant improvement of production environmental standards (e.g., F2F Strategy), often not sufficiently supported by alternative solutions in pest management. In the case of extreme and systematic weather risks, a country-wide event cannot be covered under indemnity insurance schemes because the costs for the physical damage assessment in the field often outweigh the benefit for the insured farm (Vroege et al., 2019) and the systemic nature of the event may expose insurance sector to unsustainable costs. Concerning biotic threats, the availability of insurance is rare as it is often unsuitable for the insurance market, due to both their unpredictable spread, linked to an epidemic dynamic, and agents' behavioural reasons, mainly moral hazard (Norton et al., 2016). To face the limitations of insurance, some pioneering initiatives of mutual funds to manage such risks have been locally developed in Italy, both with private and public support (Giampietri et al., 2020; Höschle et al., 2023), but the availability of such tools is below the expected demand.

To design a better-tailored RM tool offering protection against pests and diseases that are appealing to farmers, the first important step is to investigate farmers' preferences for the characteristics of such innovative tools. Until now, farmers' preferences for insurance contract characteristics remain mostly unaddressed, despite being of utmost importance for designing new insurance contracts, extending them to other crops, and increasing participation rates. To this purpose, the discrete choice experiment (DCE) approach has proved to be useful. Based on the Lancaster's (1966) theory of consumer and the Random Utility Theory, demand is defined over the characteristics of goods, rather than over goods themselves. In DCE respondents are thus asked to choose between different bundles of goods (e.g., RM tools) described in terms of their characteristics (e.g., price, level of maximum indemnity). DCE has been largely employed to elicit consumers' preferences and policy design (Colombo et al., 2005). More recently, this approach has also been used to investigate farmers' preferences for agro-environmental scheme designs (Ruto and Garrod, 2009) and contract farming configurations (Abebe et al., 2013). Furthermore, discrete choice experiments have been used to investigate farmers' preferences for insurance contract characteristics. While there are several DCE carried out in developing countries addressing farmers' preferences for insurance characteristics (Akter at al., 2016; Reynaud, Nguyen, Aubert, 2017; Ward and Makhija, 2018; Ali et al., 2021; Tang et al., 2022), there are only few that concern European farmers (Mercadé et al 2009; Liesivaara and Myyrä, 2017; Möllmann et al., 2019; Doherty et al., 2021; Čop et al., 2023) and there are no applications on Italian farmers. In conclusion, the literature review highlights the lack of a general framework to support the development of effective policies to promote RM solutions in agriculture. A comparison of different frameworks can improve the understanding of farmers' behaviour and evaluate the most suitable approach depending on the organizational profile. These studies should support policy design based on the joint study of farmers' preferences and behaviour towards RM strategies. Furthermore, there is a need to carry out policy impact assessments in terms of farmer uptake of RM innovations and effects in reducing risks and increasing farm resilience. Finally, it may prove useful, as RM solutions for farmers increase, to better understand the interactions and possible trade-offs between the different tools.

3. BEHAVIOURAL FACTORS INFLUENCING FARMERS' ADOPTION OF RISK MANAGEMENT TOOLS

From a behavioural perspective, the economic framework that is generally used to understand and predict farmers' decisions to cope with agricultural risks is rooted in expected utility theory (EUT) (von Neumann and Morgenstern, 1947). However, some fairly recent empirical applications have demonstrated that farmers' decisions to insure their production often depart from standard EUT. Non-standard economic theories, such as prospect theory, explain farmers' choice behaviour more parsimoniously (e.g., Babcock, 2015; Dalhaus et al., 2020; Feng et al., 2021). These empirical findings determined the development of a small but growing research that investigates the extent to which behavioural factors such as farmers' probabilistic beliefs, probability weighting, risk and uncertainty preferences, and loss aversion influence farmers' decisions to purchase an insurance product (e.g., Fezzi et al., 2021) and participate to mutual funds (Rippo and Cerroni, 2023; Čop et al., 2023). These behavioural factors are generally elicited using experimental methods, while their ability to explain and predict farmers' choice behaviour is tested by combining data from economic experiments with primary data obtained using stated preference surveys or available secondary data on actuarial farmers' purchasing decisions (Iver et al., 2019). In this section, we mainly focus on the literature related to subjective probabilities, risk and uncertainty preferences.

Subjective probabilities are considered to be important predictors of farmers' behaviour because farmers, like any other economic agent, base their decisions on their beliefs or expectations when the decision context is highly uncertain. If expressed in a probabilistic fashion, these beliefs or expectations are defined as subjective probabilities (e.g., Hardaker and Lien, 2010). The literature looking at the role of subjective probabilities in explaining farmers' behaviour is scant, and only a very small number of studies examined how subjective probabilities influence farmers' decision to purchase an insurance product (see Cerroni, 2020; Cop et al., 2023; Cerroni and Rippo, forthcoming for recent reviews). There are a couple of noticeable exceptions. Čop et al. (2023) found that subjective probabilities are important predictors of farmers' decisions to enrol on a sector-specific IST related to grapevine. Fezzi et al. (2021) found that farmers' subjective probabilities regarding production losses due to extreme climatic events are not in line with objective measures of risk. Hence, policy interventions geared to reduce this gap could have important policy implications regarding insurance subsidization. One potential drawback of these studies is the elicitation of subjective probabilities using hypothetical methods that are not incentive-compatible and therefore do not induce farmers to elicit truthful beliefs. However, the literature on decision analysis provides several incentive-compatible methods that are able (in theory) to elicit truthful beliefs under a proper incentive scheme. These methods could be used to elicit more accurate subjective probabilities related to uncertain agricultural outcomes (see Cerroni and Rippo, forthcoming for a review). More accurate probability assessments should have in theory a higher degree of external validity and explain farmers' choice behaviour more parsimoniously.

Risk preferences have been shown to be an important driver of farmers' decision-making processes, especially those related to the adoption of new technology and crops (e.g., Liu, 2013; Barham et al., 2016). However, only a few studies have investigated whether these preferences can play a role in explaining farmers' decision to purchase insurance products. Recent research indicates that risk preferences are poorly correlated with the decision to purchase traditional insurance products (Menapace et al., 2016; Coletta et al., 2018; Rommel et al., 2019; Cop et al., 2023). These results may be driven by some confounding factors that have been recently identified in the related literature. First, risk preferences appear to be highly context-dependent (Finger et al., 2022) and therefore their ability to explain farmers' choice behaviour may be context-dependent too. Second, a wide range of approaches exists to elicit risk preferences (see Cerroni, 2020, and Cerroni et al., forthcoming for recent reviews), and, unfortunately, empirical evidence suggests that different elicitation techniques provide inconsistent measures (e.g., Reynaud and Couture, 2012). Once again, the elicitation technique used may have an impact on the ability of elicited preferences to explain choice behaviour. Some practitioners advocate that adding an agricultural context to standard monetary lotteries can improve the external validity of elicited preferences, thus boosting the predictive power of elicited risk preferences. On the other hand, contextualization may lead farmers to use heuristics that undermine the internal validity of experimental data (see Cerroni, 2020 for an application of contextualized field experiments and a discussion on strengths and limitations).

If farmers' risk preferences are extensively researched in the related literature, uncertainty and ambiguity preferences are not. There are only very few studies eliciting farmers' uncertainty and ambiguity preferences (e.g., Beharam et al., 2014, Bougherara, 2017; Cerroni, 2020). None of these studies attempt to use such preferences to explain farmers' behaviour when purchasing insurance products. In this section, we use the terms uncertainty and ambiguity interchangeably, however, we have to acknowledge that the distinction between risk, uncertainty and ambiguity is far from being clear in the decision analysis literature (see Cerroni and Rippo, forthcoming for a discussion).

The most popular approach to disentangling these concepts is the frequentist. Here, risk refers to situations where definite numerical probabilities are known and can be objectively measured, while uncertainty refers to situations where definite numerical probabilities are unobservable (Knight, 1921). However, other paradigms exist, such as the subjectivist, under which subjective probabilities play a key role under both conditions of risk and uncertainty (Ramsey, 1931; de Finetti, 1931, Savage, 1954). Furthermore, there are other schools of thought that try to differentiate uncertainty from ambiguity. For example, according to Harrison (2011), uncertainty refers to situations when the agent can form a unique and well-defined subjective probability distribution, while ambiguity refers to situations when the agent is not capable of doing so.

This brief discussion on the role that subjective probabilities, risk and uncertainty preferences can have on farmers' decisions to use risk management tools allows to highlight a few key points relevant in the case of the use of insurances. First, the literature exploring the extent to which these behavioural factors affect these decisions is almost non-existent. The literature focusing on standard agricultural insurances and mutualistic solutions is limited. While common sense suggests that subjective probabilities, risk, and uncertainty preferences can affect the uptake of risk management tools, the extent of these impacts and the underlying behavioural mechanisms are unclear and under researched. Second, there is still an open discussion in the decision analysis literature regarding the most appropriate way to elicit these behavioural factors. Many methods are available to elicit subjective probabilities, risk and uncertainty preferences in the literature and empirical evidence suggests different methods lead to different results. This may have an influential impact on the role these behavioural factors play in explaining farmers' insurance decisions. The horse race to truthful probabilistic beliefs, risk and uncertainty preferences is not over yet. Exploring the internal and external validity of results obtained via different elicitation methods appears to be the only strategy to shed light on these issues. Third, behavioural factors can be useful also to predict farmers' choice behaviour. A new stream of research is emerging that seeks to incorporate these behavioural factors into simulation models to enhance their ability to explain and predict choice behaviour (e.g., Huber et al., 2022). This line of research definitively contributes to build a more holistic view about. sustainable risk management in agriculture.

4. INNOVATIVE RISK MANAGEMENT TOOLS

The previous sections have focused on RM behaviour and choices of RM tools mainly referring to already existing tools such as crop insurance schemes that have a long history in Italy, and date back to the early 2000s (Cafiero et al., 2007; Santeramo and Ramsey, 2017). The transition has been motivated by drawbacks associated with the ex-post compensation, such as its financial unsustainability (Goodwin and Smith, 1995; Goodwin and Mahul, 2004; Mahul and Stutley, 2010; Santeramo et al., 2016), as compared to crop insurance and revenue insurance schemes.

Despite this, the public crop insurance scheme has not been a story of success, as testified by low and heterogeneous participation and retention (Santeramo et al., 2016). This has been motivated by the lack of tradition with subsidized schemes, as well due to the necessity to serve a relatively little market, with many (highly differentiated) crops and a majority of small-size firms (Santeramo et al, 2016) and has suggested to implement ameliorative reforms to overcome the complexity of the policy environment (Severini et al., 2017). Two major reforms were implemented in 2013 and 2015; the former removed subsidies to the mono-risk insurance contracts; the latter replaced the multi- and pluri-risks contract schemes with "packages" covering a set of adversities (Santeramo et al., 2022). Both reforms had negligible effects on insured acreage (as high as three percent) and insured values (estimated to be lower than one percent), casting doubts on their effectiveness. Finally, farmers are often coping with risks which are not covered by traditional insurance schemes, or that are extraordinary in terms of expected damages (i.e., due to so called catastrophic events).

Because of these reasons, it seems relevant to consider innovative RM tools because these may overcome the issues encountered by the traditional insurance products. Indeed, the new Common Agricultural Policy is continuing to reform by enlarging the support for innovative risk management interventions and strategies. Here we focus on two of themes: mutual funds for the so-called catastrophic events and the index-based insurance schemes. These options are precisely meant to complement traditional insurance schemes but are not the only innovative instruments¹. While the ex-post approach tries to limit the potential additional damages that may occur after a catastrophic event, and to promote the restoration of the damaged structures, alternative mechanisms may help share the costs associated with extreme events and catastrophes. The use of insurance tools to cope with extreme and catastrophic events is dated (Michel-Kerjan, 2010) but still very debated, especially in agriculture (Bucheli et al., 2020). The rationale is simple: due to the increasing amount of available data on weather conditions and the higher frequency of extreme events and natural disasters (both systemic in nature and with high impacts on the sector), coping more directly with these events is not only possible, but also necessary to avoid the default of many farms. The catastrophic bonds and the catastrophic reinsurance may help cope with disasters as they are bet on the occurrence of a disaster, in which case an indemnity is paid. Another possibility, being explored in Italy, is the use of a mutual fund to cover losses from high impactful events.

In Italy the new risk management interventions have been defined by the National Strategic Plan (NSP) 2023-2027 of the CAP. Besides confirming support to (production) insurance schemes, (production) mutual funds (for plants, animal production, farm structures and livestock farms), and (income) mutual funds (for selected sectors such as poultry, sugar beet, durum wheat, cow and sheep milk, olive, fruit and vegetable, rice, and pig), it also establishes a mutual fund (the Agricat) for catastrophic event. The latter covers farmers against specific weather events (frost, drought, and flood) defined as potentially catastrophic. The indemnities are triggered by production losses due to one of the three events, as certified by randomly executed expert reports. The economic sustainability of the newly established fund is unclear, and depends on the design of the fund, on the rating of the premia, and on the effectiveness of the damage reports. In 2022 the Ministry of Agriculture has started a pilot study in thirteen provinces (both in the North and in South), for twelve products (apples, pears, durum wheat, corn, almonds, oranges, apricots, actinidia, wine grapes, oil olives, peaches, and industrial tomatoes). ISMEA has released preliminary results on peaches, concluding that about sixty percent of insured farms have incurred in losses below the (20%) damage or (30%) deductible thresholds, whereas the remaining share of farms has incurred in losses as high as fifty percent in one out of four cases. The findings are of undoubted interest, but also worrisome compared to the US system, in which indemnities worth 14.9% of insured liability are considered to be excessively high in that the median value is as low as 2% of crop insurance liabilities (DeLay et al., 2022). Such a high level of indemnities points at precise future goals, which can be summarised in four priorities: i) improve the accuracy of existing data to map and monitor high-impacts weather events; ii) increase the penetration of the program, and farm retention; iii) mitigate and reduce repetitive losses to lower operating expenses; iv) strengthen the financial sustainability of the program by designing optimal participation rates. These points translate into research questions worth investigation.

Another innovation, barely adopted in developed countries, is the use of index-based insurances: the

¹ For sake of brevity, we do not discuss other innovative instruments such as, for instance, the Income Stabilization Tool (cfr. Giampietri et al., 2020, and Zinnanti et al., 2022, for recent assessments).

scheme indemnifies farmers, who have likely incurred losses, when the index exceeds a threshold (Abdi et al., 2022). A practical example may help in understanding the rationale behind this scheme. Assume a set of farms are exposed to potentially detrimental events (e.g., excessive rain): when the event occurs with a certain magnitude (e.g., the daily volume of precipitation is three times larger than the average precipitation) the likelihood that the farm experiences a loss is high. In this situation, while collecting data on precipitation for each single farm may be costly and inappropriate to determine the incurred losses, relying on the index may be a second-best solution. The operation of an index-based insurance consists of indemnifying all farms when the index exceeds a threshold. The scheme of an index-based insurance has pros and cons, and is not the ultimate solution (Carter et al., 2017): on one hand it may reduce (or eliminate) moral hazard and adverse selection issues; on the other hand, it may be ineffective if the correlation between triggered pay-outs and the occurrence of loss events is rather low. This potential fallacy is referred to as "basis risk" and defined as "the risk that a protection buyer's own losses exceed the payments under a risk transfer mechanism structured to hedge against these losses." (Ross and Williams, 2009). The basis risk has serious impacts on the functioning of index insurance (Clement at el., 2020) and calls for a deep understanding of the phenomena aimed to be coped against.

The basis risk is a three-dimensional concept, defined by time, space and design of the index. The three dimensions correspond to the temporal, spatial and design basis risks, which are inversely correlated with the informative content of the information being used. For instance, an index-insurance built on hourly data for weather events provides lower temporal basis risks with respect to the same index built on annual data. Similarly, an index built on state level data will have higher spatial basis risk than a similar index relying on municipality level data. As for the design basis risk, the lower the flexibility (and complexity) of the index, the higher the design basis is likely to be. Differently, highly informative datasets allow good performance, in terms of correlation between agricultural data and weather statistics (e.g., Cheng et al., 2017). In a recent paper, Stigler and Lobell (2023) decompose, from a theoretical point of view, the basis risk of the index insurance schemes. Their empirical analysis uses linear and quantile regressions, coupled with richly informative datasets, to derive effective indexes.

Index-based insurance is still underutilized in developed countries, whereas there are several applications in developing countries: "more than fifteen developing countries have offered individual-level index insurance schemes [...], and some twenty have offered it at the institutional or geographical level." (Carter et al., 2017, p. 423). The low uptake of index-based insurance calls for a better understanding of the challenges that prevent participation. In particular, Carter et al. (2017) indicate four areas of improvement on (a) the design of the contract; (b) the measurement of risks; (c) the quality of insurance schemes, and (d) the use of other risk coping interventions.

These areas of improvement should be approached by promoting empirical studies that explore the informative content of the large datasets, through a holistic lens capable of merging knowledge from different disciplines (e.g. climatology, agronomy, statistics, economics, management) . For instance, promising research may be conducted by analysing the correlation in the tails of the yield and weather data distribution² (e.g., copula-based models, quantile regressions) as well as using quantitative methods capable of synthesising large sets of variables³ (e.g., machine learning, shrinkage estimators). However, none of those techniques would be sufficient without a better understanding of the fundamentals of the economics and management of risks, topics that should remain a priority in the research agenda.

5. AGRICULTURAL RISK, INPUT USE AND RELATED POLICIES

The previous sections have considered the potential role of RM strategies and tools without accounting for the fact that other policies exist and affect farmers' behaviour. Indeed, some policies not targeted to risk in agriculture may have an unintended effect on RM behaviour and choices. A holistic view to the risk analysis cannot ignore the synergies and trade-off across policies directly or indirectly affecting the agricultural risk. This section refers to the literature that has shown that there exists a relation between the farmer's use of fertilisers and pesticides and the risk farmer faces. Thus, any policy aimed at constraining the use of chemicals in agriculture has an indirect effect on the risk the farmers must cope with.

This branch of analysis fits with the EU F2F Strategy, which has defined a set of objectives and guidelines to drive the European agri-food system toward a fair,

² See for instance applications by Goodwin and Hungerford (2015), Conradt et al. (2015), Bokusheva (2018), among others.

³ A good discussion on recent methodological advances to model insurance is provided by Ali et al. (2020) who discuss the potentiality of machine learning, as well as of artificial intelligence.

healthy, and environmental-friendly transition. Among the targets of the Strategy, two are specifically addressed to fertilisers and pesticide use. The Strategy aims at reducing the overall application of chemical pesticides in agriculture by 50% as well as of the more hazardous ones by 50% by 2030. The Strategy also envisages a reduction in fertiliser application of 20% by 2030 and a reduction of the nutrient losses in the soil of 50% over the same time horizon. Along this line, in June 2022 the Commission made a proposal (European Commission, 2022) to revise the directive on the sustainable use of pesticides (2009/128/EC) and to switch the legal framework from a directive to a regulation. The proposal aims at meeting the F2F Strategy targets in terms of pesticides and has received criticism from many Member States which particularly blame the "flat rate" pesticide cut proposed by the Commission. In order to implement the fertiliser goal of the Strategy, a new digital tool is being developed, the Farm Sustainability Tool for Nutrients (FaST). FaST will combine data from different sources and will provide detailed ad hoc recommendations on the application of crop fertilisation and plant protection products. This should improve the efficiency in fertiliser use and, consequently, should help to comply with the target on the fertiliser use reduction and losses. Finally, measures related to the pesticides and fertilisers targets are contained in the National Strategic Plans of the Common Agricultural Policy (CAP) 2023-2027, specifically in the eco-schemes and in the enhanced conditionality of the first CAP pillar as well as in the agrienvironmental-climate measures of the second CAP pillar. The effect of a policy targeted at the reduction of chemicals is likely to affect farm input decisions and the corresponding farmers' expected utility in two ways. First, it would directly constrain the amount of chemical application in the farmer's decision process. Second, as literature has shown that pesticides and fertilisers use often changes the level of agricultural risk and farmers are usually risk-averse, the policy imposed on these inputs alters the level of risk the farmers face and, as a consequence, is likely to affect farmer's decision on the use of other inputs related to the chemicals (may they be substitutes or complementary inputs).

The use of agricultural inputs, including pesticides and fertilisers, is affected by the degree of risk aversion of the farmer (Bontemps et al., 2021). For example, according to the model adopted (either the Expected Utility or the Cumulative Prospect Theory (Tversky and Kahneman, 1992)) it has been shown that from 4% to 19% of the pesticide expenditure on farms is explained by farmer risk aversion behaviour (Bontemps et al., 2021). Therefore, the higher the risk aversion of the farmers the higher will be the impact on farmer's decisions and, in turn, on the effectiveness of a policy that restricts chemicals use. The estimation of the farm risk aversion behaviour is partially dependent on the theoretical model adopted. For instance, Rommel et al. (2022) in a study on farmer's risk preferences in 11 EU countries have shown that the Cumulative Prospect Theory explains the preferences better compared to the Expected Utility framework.

Möhring et al. (2020a) provide a deep literature review on the relationship between pesticide use and farm risk. They show that around half of the papers that so far have assessed this issue find a risk-increasing effect of the pesticides (e.g., Serra et al., 2006 and 2008), around half report a risk-decreasing effect (e.g., Koundouri et al., 2009; Antle, 2010; Gardebroek et al., 2010) and only one paper finds no effect (Hurd, 1994). Besides the heterogeneity of the agricultural products and of the countries analysed in the papers, another reason to explain such opposite effects found in the literature is the heterogeneity of the indicators used to measure pesticide application across papers. Möhring et al. (2020a) show that the impact of pesticide use on farm risk depends on the pesticide indicators and on the pesticide type considered. Unfortunately, in most of the countries the national FADN datasets contain data of poor quality related to pesticide quantity and this prevents from using that type of information in this type of research. If no better-quality data are available, the only option is to use pesticide expenditure as a proxy for pesticide quantity. Indeed, most of the studies cited above use pesticide expenditure. Improving the data quality concerning pesticides in the EU is of paramount importance to inform evidence-based policy making. The F2F Strategy also acknowledges the need to overcome the data gaps by changing the 2009 Regulation concerning statistics on pesticides. In addition, the Commission announced its intention to convert the FADN dataset into the FSDN (Farm Sustainability Data Network) which would include more detailed information on the environmental practices of the farms and would introduce data on their social practices (European Commission, 2021).

When it comes to fertilisers, literature (Paulson and Babcock, 2010; SriRamaratnam et al., 1987) agrees that fertilisers are risk-increasing and it highlights the "fertilisers paradox". Indeed, although fertilisers are riskincreasing, risk averse farmers oversupply them. This happens because, under production uncertainty, due to for example unpredictable weather conditions or uncertainty in the amount of nutrients available to the crops, the overapplication of fertilisers is used by farmers as a form of self-protection (Babcock, 2001; Paulson and Babcock, 2010). Therefore, when modelling farmer fertiliser use it is important to account for the uncertain conditions the agricultural production faces which affect farmer's decisions about fertilisers.

When analysing the relationship between chemical use and agricultural risk some issues must be addressed from a methodological viewpoint. First, there may be some simultaneity in the chemical use decision and agricultural risk. Indeed, usually chemicals are applied multiple times in the cropping seasons and the number of applications as well as the amount of chemicals applied in each application are decided by the farmers based on how the season is going in terms of weather and pests and how the crop is growing. If throughout the cropping season, the farmer observes an increase in agricultural risk compared to his initial expectation, he may adjust the planned fertiliser and pesticide application consequently. In addition, the past year agricultural risk is likely to affect the current year input decisions. This consideration makes relevant the use of a sequential decision-making production model as proposed by Antle (1983a) to account for the possible feedback effect, i.e., the farmer adjusts his decision on variable input use based on the output observed or on the adjustment in the output expectations. Second, the analysis of risk in agriculture needs to account not only for the mean and variance of the crop yields and of farmer's revenue, but also for higher moments of the yield and revenue distribution (Finger et al., 2018) as farmers are often downside risk averse (Di Falco and Chavas, 2006). Indeed, it is likely that farmer decisions are more affected by variations of crop yield and farm revenue below the average than by variations above the average. Specific econometrics approaches exist to address higher moments of the distribution such as the moment-based approach (Antle, 1983b) recently applied to Italian farm data by Biagini et al. (2022b) and its updated version that uses partial moments (Antle, 2010).

Another important topic only partially addressed in the literature is the relationship between RM tools adoption and chemical application. RM tools change the agricultural risk faced by farmers and consequently, they may impact the use of risk-increasing and riskdecreasing input when the farmer is not risk-neutral. For example, the adoption of insurance may induce farmers to adopt moral hazard behaviour which in turn influences the pesticide and fertiliser use decisions (Mishra, 2005). As stated earlier, pesticides are risk-increasing or risk-decreasing according to the type of crop, the country and the pesticide indicator considered, and therefore the direction of the relationship between pesticides and farm insurance remains an empirical question. Second, insurance may induce farmers to change land allocation among crops, e.g. by growing more risky crops, and this also affects chemical use. Analysing whether insurance and chemicals are substitutes, complements or independent goods from the producer's perspective is a worthy issue. Indeed, this assessment would outline the interaction and the spillover effects among two policies: the one pointing at reducing chemicals application in agriculture and the one pointing at increasing the adoption of RM tools among farmers. Although this research branch is rather explored in the US where studies reveal a positive, a negative and a zero effect of insurance uptake on input use, Möhring et al. (2020b) is one of the few examples applied to European agriculture. The study is focused on two countries (France and Switzerland) and it shows that the adoption of insurances increases the application of pesticides.

6. CONCLUSIONS

The agricultural sector is, by its nature, exposed to several risks, and farmers have a long-lasting tradition in coping with them. Yet the rising complexity and interconnection of the global agri-food sector have introduced new risks into the sector, calling for more and more frequent policy interventions. In the European Union, the interventions on risk management have been fragmented and managed at the national level and slow is the introduction of innovative tools. In the context of a CAP with a declining budget, oriented toward environmentally-friendly and sustainable production models, risk management becomes a relevant strategy to lower income uncertainty and favour the resilience of the agri-food system. Given the low uptake rate of RM tools in the European Union, studying the determinants of the uptake referring to both the farm and farmer's characteristics as well as to the RM tool design is crucial to set up effective innovative RM tools and to refine the traditional ones. Studies on the determinants of farmers' behaviour towards risk management tools are rather limited in number. Equally, literature on innovative tools, such as index-based and the catastrophic insurance schemes is scant.

The role of behavioural factors in explaining farmers' risk management decisions is often neglected in the literature. However recent studies suggest that behavioural factors should be incorporated into models to explain and predict farmers' adoption and use of risk management tools (e.g., Babcock, 2015; Dalhaus et al., 2020; Feng et al., 2020; Tack and Yu, 2021). Those studies have stimulated a growing stream of research that has mainly focused on the elicitation of farmers' risk preferences using experimental methods. Empirical results show that further research is needed in several dimensions. First, preference elicitation methods appear to have an important impact on the magnitude of elicited preferences, therefore further research is needed to identify the methods that provide more robust measures in terms of internal and external validity. Second, empirical evidence indicates that risk preferences are context-dependent. Hence, it is still unclear whether risk preferences elicited using monetary lotteries are fully able to explain farmers' choice behaviour, or whether practitioners should move to the use of contextualized lotteries that may improve the external validity of elicited preferences. Third, while farmers' risk preferences have been widely investigated in the literature, there are only a few studies focusing on the role that subjective beliefs, ambiguity attitudes and time preferences may have on farmers' decisions in general, and more specifically, regarding the uptake of different RM tools. Finally, a holistic approach to the study of risk management cannot ignore the interrelation between farmer's input not specifically targeted to risk management and the agricultural risk. This is key to discover possible synergies and trade-offs among different policies. The policies aiming at reducing the fertiliser and pesticide use in agriculture indirectly affect the agricultural risk because fertilisers and pesticide impact the risk level. Results reported by the literature on this impact are controversial and depend on the crop, country and indicator considered and there is often the issue of poor data quality. A more sustainable use of chemicals in agriculture and a better management of risk in the sector are two forefront topics in agricultural economics. Hence, efforts towards gathering better quality data are required. In addition, given the importance of framing consistent and effective EU policies, the investigation of the relationship between the RM tools adoption and the farm application of chemicals needs to be addressed. This investigation would shed light on whether two apparently independent policies, namely the policy restricting the use of chemicals in agriculture and the one promoting the adoption of RM tools, have the same or opposite direction.

Besides briefly mentioning what the literature has already provided and which methods have been investigated, we conclude this paper with a few priorities to orient future research: efforts should be devoted to improve the use of the large amount of available data, to improve the financial mechanisms that may ensure the financial stability of the RM schemes, and to increase the interconnection (and complementarity) of the RM instruments. In short, the search for innovation in RM should be not only oriented toward a sophistication of the strategies, but also (and mainly) toward a better exploitation of the informative content of the existing data, as well as of the holistic nature of the approaches. In addition, the results of this literature review suggests that the theoretical framework used so far in the literature for understanding farmer behaviour in terms of RM is not unique: different models have been applied sparsely, and often without accounting for the complex nature of the issue at stake. Therefore, it seems necessary to commit research efforts to carrying out a comparative evaluation of methods and hypotheses used in empirical analyses, also including behavioural variables towards risk. This latter calls for developing new tools to investigate farmers' preferences, with particular attention to the characteristics of the tools and their interaction with other strategies. Similarly, because farmers' behaviour is affected by several policies, including those aimed at reducing the use of potentially harmful inputs, it seems important to analyse RM choices under a more articulated policy scenario. Fostering the analyses in these directions is expected to better understand how farmers select among available RM instruments including the most innovative ones on which they have not a large experience. The results of these analyses could provide insights that could be used to increase the uptake of already existing RM tools, facilitate the design and introduction of the new ones and, indirectly, allow the EU farm sector to become more resilient.

ACKNOWLEDGEMENT

This work has been developed within the PRIN project 20205L79R9 funded by the Italian Ministry of University and Research (MUR), and within the Agritech National Research Center and received funding from the European Union Next-GenerationEU (PIANO NAZIONALE DI RIPRESA E RESILIENZA (PNRR) – MISSIONE 4 COMPONENTE 2, INVESTIMENTO 1.4 – D.D. 1032 17/06/2022, CN00000022). This manuscript reflects only the authors' views and opinions; neither the European Union nor the European Commission can be considered responsible for them.

REFERENCES

Abdi, M. J., Raffar, N., Zulkafli, Z., Nurulhuda, K., Rehan, B. M., Muharam, F. M., Khosim, N.A., and Tangang,

F. (2022). Index-based insurance and hydroclimatic risk management in agriculture: A systematic review of index selection and yield-index modelling methods. International Journal of Disaster Risk Reduction, 67: 102653.

- Abebe, G. K., Bijman, J., Kemp, R., Omta, O., and Tsegaye, A. (2013). Contract farming configuration: Smallholders' preferences for contract design attributes. Food Policy, 40: 14-24.
- Ajzen, I. (1985), From Intentions to Actions: A Theory of Planned Behavior, Springer, Berlin.
- Ajzen, I. (1991). The theory of planned behavior, Organizational Behavior and Human Decision Processes, 50: 179-211.
- Akter, S., Krupnik, T. J., Rossi, F., and Khanam, F. (2016). The influence of gender and product design on farmers' preferences for weather-indexed crop insurance. Global Environmental Change, 38: 217-229.
- Ali, W., Abdulai, A., and Mishra, A. K. (2020). Recent advances in the analyses of demand for agricultural insurance in developing and emerging countries. Annual Review of Resource Economics, 12: 411-430.
- Ali, W., Abdulai, A., Goetz, R., and Owusu, V. (2021). Risk, ambiguity and willingness to participate in crop insurance programs: Evidence from a field experiment. Australian Journal of Agricultural and Resource Economics, 65(3): 679-703.
- Antle, J. M. (1983a). Sequential decision making in production models. American Journal of Agricultural Economics, 65(2): 282-290.
- Antle, J. M. (1983b). Testing the stochastic structure of production: a flexible moment-based approach. Journal of Business & Economic Statistics, 1(3): 192-201.
- Antle, J. M. (2010). Asymmetry, partial moments and production risk. American Journal of Agricultural Economics, 92(5): 1294–1309.
- Babcock, B. A. (1992). The Effects of Uncertainty on Optimal Nitrogen Applications. Review of Agricultural Economics. 14(2): 271–280.Babcock, B.A. (2015). Using Cumulative Prospect Theory to Explain Anomalous Crop Insurance Coverage Choice. American Journal of Agricultural Economics, 97(5): 1371-1384.
- Barham, B.L., Chavas, J.-P., Fitz, D., Salas, V.R., and Schechter, L. (2014) The roles of risk and ambiguity in technology adoption. Journal of Economic Behavior & Organization, 97: 204–218. https://doi. org/10.1016/j.jebo.2013.06.014
- Biagini, L., and Severini, S. (2022a). Can Machine Learning discover the determining factors in participation in insurance schemes? A comparative analysis. Working Paper, December 2022. DOI: 10.48550/arXiv.2212.03092

- Biagini, L., and Severini, S. (2022b). How Does the Farmer Strike a Balance between Income and Risk across Inputs? An Application in Italian Field Crop Farms. Sustainability (Switzerland), 14(23). https://doi. org/10.3390/su142316098
- Bokusheva, R. (2018). Using copulas for rating weather index insurance contracts. Journal of Applied Statistics, 45(13): 2328-2356.
- Bontemps, C., Bougherara, D., and Nauges, C. (2021). Do risk preferences really matter? the case of pesticide use in agriculture. Environmental Modeling & Assessment, 26(4): 609-630.
- Bougherara, D., Gassmann, X., Piet, L., and Reynaud, A. (2017). Structural estimation of farmers' risk and ambiguity preferences: a field experiment. European Review of Agricultural Economics, 44(5): 782-808.
- Bucheli, J., Dalhaus, T., and Finger, R. (2021) The optimal drought index for designing weather index insurance. European Review of Agricultural Economics, 48(3): 573-597.
- Cafiero, C., Capitanio, F., Cioffi, A., and Coppola, A. (2007). Risk and crisis management in the reformed European agricultural policy. Canadian Journal of Agricultural Economics/Revue canadienne d'agroeconomie, 55(4): 419-441.
- Capitanio, F., Adinolfi, F., and Di Pasquale, J. (2016). The income stabilization tool: Assessing the hypothesis of a national mutual fund in Italy, American Journal of Applied Sciences, 13 (4): 357-363
- Carter, M., de Janvry, A., Sadoulet, E., and Sarris, A. (2017). Index insurance for developing country agriculture: a reassessment. Annual Review of Resource Economics, 9(1): 421-38.
- Cerroni, S. (2020). Eliciting farmers' subjective probabilities, risk, and uncertainty preferences using contextualized field experiments. Agricultural Economics, 51: 707–724. https://doi.org/10.1111/agec.12587
- Cerroni, S., and Rippo R. (Forthcoming). Subjective probabilities and farmers' decision-making in developing countries. In: Mishra, A., Kumbhakar, S.C., Lien, G. (Eds.) Managing risk in Agriculture: A Development Perspective. CABI books.
- Cerroni S., Bozzola M., and Rippo R. (Forthcoming). Eliciting farmers' risk and uncertainty preferences using contextualized field experiments in developing countries: A pilot study. In: Mishra, A., Kumbhakar, S.C., Lien, G. (Eds.) Managing risk in Agriculture: A Development Perspective. CABI books.
- Chartier, O., Cronin, E., Jongeneel, M., van Asseldonk, M., Bergevoet, R., Cordier, J., Gielen-Meuwissen, M. and Bocci, M. (2017). Study on Risk Management in EU Agriculture. (DG AGRI European Commission: Brussels).

- Chen, W., Hohl, R., and Tiong, L. K. (2017). Rainfall index insurance for corn farmers in Shandong based on high-resolution weather and yield data. Agricultural Finance Review, 7(2): 337-354.
- Clement, K. Y., Botzen, W. W., Brouwer, R., and Aerts, J. C. (2018). A global review of the impact of basis risk on the functioning of and demand for index insurance. International Journal of Disaster Risk Reduction, 28: 845-853.
- Coletta, A., Giampietri, E., Santeramo, F. G., Severini, S., and Trestini, S. (2018). A preliminary test on risk and ambiguity attitudes, and time preferences in decisions under uncertainty: towards a better explanation of participation in crop insurance schemes. Bio-based and Applied Economics, 7(3): 265-277.
- Colombo, S., Hanley, N., and Calatrava-Requena, J. (2005). Designing policy for reducing the off-farm effects of soil erosion using choice experiments. Journal of Agricultural Economics, 56(1): 81-95.
- Conradt, S., Finger, R., and Bokusheva, R. (2015). Tailored to the extremes: Quantile regression for indexbased insurance contract design. Agricultural economics, 46(4): 537-547.
- Čop, T, Cerroni, S., and Njavro, M., 2023. Farmers' acceptance of the income stabilisation tool: a discrete choice experiment application. European Review of Agricultural Economics, 50(4): 1520–1546.
- Cordier, J., and Santeramo, F. (2020). Mutual funds and the Income Stabilisation Tool in the EU: Retrospect and Prospects. EuroChoices, 19(1): 53-58.
- Dalhaus, T., B.J. Barnett, and R. Finger. (2020). Behavioral weather insurance: Applying cumulative prospect theory to agricultural insurance design under narrow framing" I. Rodriguez-Lara, ed. PLOS ONE, 15(5): 1-25.
- Davis, F.D. (1986). A technology acceptance model for empirically testing new end-user information systems: theory and results, PhD thesis, Sloan School of Management, Massachusetts Institute of Technology.
- de Finetti, B. (1931a). Sul significato soggettivo della probabilità. Fundamenta mathematicae 17, 298-329. Translated into English as On the subjective meaning of probability, in Monari, P. and Cocchi, D. (eds.), Probability and Induction, Bologna, Clueb: 291-321
- DeLay, N. D., Brewer, B., Featherstone, A., and Boussios, D. (2022). The impact of crop insurance on farm financial outcomes. Applied Economic Perspectives and Policy, 45(1): 579-601.
- Di Falco, S., and Chavas, J. P. (2006). Crop genetic diversity, farm productivity and the management of environmental risk in rainfed agriculture. European review of agricultural economics, 33(3): 289-314.

- Doherty, E., Mellett, S., Norton, D., McDermott, T. K., O'Hora, D., and Ryan, M. (2021). A discrete choice experiment exploring farmer preferences for insurance against extreme weather events. Journal of Environmental Management, 290, 112607.
- Dohmen, T., Falk, A., Huffman, D., Sunde, U., Schupp, J., and Wagner, G. G. (2011). Individual risk attitudes: Measurement, determinants, and behavioral consequences. Journal of the European Economic Association, 9(3): 522-550.
- Donkers, B., Melenberg, B., and Soest, A.V., (2001). Estimating risk attitudes using lotteries: A large sample approach. J. Risk Uncert. 22, 165–195.
- Enjolras, G., and Sentis, P. (2011) Crop insurance policies and purchases in France. Agricultural Economics, 42(4): 475–486.
- European Commission (2020). A Farm to Fork Strategy for a fair, healthy and environmentally-friendly food system. https://eur-lex.europa.eu/resource. html?uri=cellar:ea0f9f73-9ab2-11ea-9d2d-01aa75ed71a1.0001.02/DOC_1&format=PDF. Accessed on 14th October 2022
- European Commission (2021). Farm Accountancy Data Network. https://agriculture.ec.europa.eu/data-andanalysis/farm-structures-and-economics/fadn_en Accessed on 23rd January 2023
- European Commission (2022). Proposal for a regulation of the European Parliament and of the Council on the sustainable use of plant protection products. https://eur-lex.europa.eu/resource. html?uri=cellar:78120cfb-f5e4-11ec-b976-01aa75ed71a1.0001.02/DOC_1&format=PDF. Accessed on 17th October 2022
- European Parliament (2022). Russia's war on Ukraine: Impact on global food security and EU response. Briefing 29-09-2022. Available at: https://www. europarl.europa.eu/thinktank/en/document/EPRS_ BRI(2022)733667#:~:text=Besides%20huge%20 human%20losses%20and,and%20the%20COVID%20 19%20pandemic. Accessed on: 09/11/2022.
- European Parliament and Council (2009). Directive 2009/128/EC establishing a framework for Community action to achieve the sustainable use of pesticides. https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:02009L0128-20091125&from=EN. Accessed on 7th November 2022.
- European Parliament and Council (2009). Regulation No 1185/2009 concerning statistics on pesticides. https:// eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri =CELEX:32009R1185&from=EN. Accessed on 19th October 2022.

- Feng, H., Du, X., and Hennessy, D. A. (2020). Depressed demand for crop insurance contracts, and a rationale based on third generation Prospect Theory. Agricultural Economics, 51(1): 59-73.
- Fezzi, C., Menapace, L., and Raffaelli, R. (2021). Estimating risk preferences integrating insurance choices with subjective beliefs. European Economic Review, 135, 103717.
- Finger R., Vroege W., Spiegel A., de Mey Y., Slijper T., Poortvliet P.M., Urquhart J., Vigani M., Nicholas-Davies P., Soriano B., Garrido A., Severini S., Meuwissen, M.P.M. (2022). "The Importance of Improving and Enlarging the Scope of Risk Management to Enhance Resilience in European Agriculture". In Meuwissen, M., Feindt, P., Garrido, A., Mathijs, E., Soriano, B., Urquhart, J. et al. (Eds.) (2022). Resilient and Sustainable Farming Systems in Europe: Exploring Diversity and Pathways (Chapter 2, pp. 18-37). Cambridge: Cambridge University Press. doi:10.1017/9781009093569.003
- Finger, R., Dalhaus, T., Allendorf, J., and Hirsch, S. (2018). Determinants of downside risk exposure of dairy farms. European Review of Agricultural Economics, 45(4): 641-674.
- Franken, J. R., Pennings, J. M., and Garcia, P. (2017). Risk attitudes and the structure of decision-making: evidence from the Illinois hog industry. Agricultural economics, 48(1), 41-50.
- Gardebroek, C., Chavez, M. D., and Oude Lansink, A. (2010). Analysing production technology and risk in organic and conventional Dutch arable farming using panel data. Journal of Agricultural Economics, 61(1): 60–75.
- Giampietri, E., Yu, X., and Trestini, S. (2020). The role of trust and perceived barriers on farmer's intention to adopt risk management tools. Bio-based and Applied Economics, 9(1): 1-24.
- Goodwin, B. K., and Hungerford, A. (2015). Copulabased models of systemic risk in US agriculture: Implications for crop insurance and reinsurance contracts. American Journal of Agricultural Economics, 97(3): 879-896.
- Goodwin, B. K., and Mahul, O. (2004). Risk modeling concepts relating to the design and rating of agricultural insurance contracts (Vol. 3392). World Bank Publications.
- Goodwin, B. K., and Smith, V. H. (1995). The economics of crop insurance and disaster aid. American Enterprise Institute.
- Hardaker, J.B., and Lien, G. (2010). Probabilities for decision analysis in agriculture and rural resource economics: The need for a paradigm change. Agricul-

tural Systems, 103: 345–350. https://doi.org/10.1016/j. agsy.2010.01.001

- Harrison, G. W. (2011). Experimental methods and the welfare evaluation of policy lotteries. European Review of Agricultural Economics, 38(3): 335–360.
- Harrison, G.W., and Ng, J.M. (2019). Behavioral insurance and economic theory: a literature review. Risk Management and Insurance Review, 22(2): 133–182.
- Holt, C., and S. Laury (2002). Risk Aversion and Incentive Effects. American Economic Review, 92: 1644– 1655.
- Höschle, L., Trestini, S., and Giampietri, E. (2023). Participation in a mutual fund covering losses due to pest infestation: analyzing key predictors of farmers' interest through machine learning (accepted by International Food and Agribusiness Management Review)
- Huber, R., Xiong, H., Keller, K., Finger, R. (2022). Bridging behavioural factors and standard bio-economic modelling in an agent-based modelling framework. Journal of Agricultural Economics 73: 35–63. https:// doi.org/10.1111/1477-9552.12447
- Hurd, B. H. (1994). Yield response and production risk: An analysis of integrated pest management in cotton. Journal of Agricultural and Resource Economics, 19(2): 313–326.
- Ismea (2022). Rapporto ISMEA sulla gestione del rischio in Agricoltura 2022, ISMEA, Roma. Available at: https://www.ismea.it/flex/cm/pages/ServeBLOB.php/L/ IT/IDPagina/11842 (Accessed 05 January 2023).
- Iyer, P., Bozzola, M., Hirsch, S., Meraner, M., and Finger, R. (2020). Measuring Farmer Risk Preferences in Europe: A Systematic Review. Journal of Agricultural Economics, 71: 3–26. https://doi.org/10.1111/1477-9552.12325
- Rommel, J., Sagebiel, J., Baaken, M. C., Barreiro-Hurlé, J., Bougherara, D., Cembalo, L., Cerjak, M., Čop, T., Czajkowski, M., Espinosa-Goded, M. et al. (2012). Farmers' risk preferences in 11 European farming systems: A multi-country replication of Bocquého et al. (2014). Applied Economic Perspectives and Policy, Early View. https://doi.org/10.1002/aepp.13330
- Jensen, N.D., Mude, A.G. and Barrett, C.B. (2018). How basis risk and spatiotemporal adverse selection influence demand for index insurance: evidence from northern Kenya. Food Policy, 74: 172–198.
- Kline, R. B. (1998). Principles and practice of structural equation modeling. New York: Guilford
- Knight, F.H. (1921). Risk, Uncertainty and Profit. University of Illinois, Urbana-Champaign's Academy for Entrepreneurial Leadership Historical Research Reference in Entrepreneurship.
- Komarek, A.M., De Pinto, A., and Smith, V.H. (2020) A review of types of risks in agriculture: What we know

and what we need to know. Agricultural Systems 178, 102738. https://doi.org/10.1016/j.agsy.2019.102738

- Koundouri, P., Laukkanen, M., Myyrä, S., and Nauges, C. (2009). The effects of EU agricultural policy changes on farmers' risk attitudes. European Review of Agricultural Economics, 36(1): 53-77.
- Lancaster, K. J. (1966). A new approach to consumer theory. Journal of political economy, 74(2): 132-157.
- Liesivaara, P., and Myyrä, S. (2017). The demand for public-private crop insurance and government disaster relief. Journal of Policy Modeling, 39(1): 19-34.
- Liu, E.M. (2013). Time to Change What to Sow: Risk Preferences and Technology Adoption Decisions of Cotton Farmers in China. The Review of Economics and Statistics 95, 1386–1403. https://doi.org/10.1162/ REST_a_00295
- Mahul, O., and Stutley, C. J. (2010). Government support to agricultural insurance: challenges and options for developing countries. World Bank Publications.
- Mare, C., Manaţe, D., Mureşan, G. M., Dragoş, S. L., Dragoş, C. M., and Purcel, A. A. (2022). Machine Learning Models for Predicting Romanian Farmers' Purchase of Crop Insurance. Mathematics, 10(19): 3625.
- Menapace, L., Colson, G., and Raffaelli, R. (2013). Risk aversion, subjective beliefs, and farmer risk management strategies. American Journal of Agricultural Economics, 95(2): 384-389.
- Menapace, L., Colson, G., and Raffaelli, R. (2016). A comparison of hypothetical risk attitude elicitation instruments for explaining farmer crop insurance purchases. European Review of Agricultural Economics, 43: 113–135.
- Mercadé, L., Gil Roig, J. M., Kallas, Z., and Serra, J. (2009). A choice experiment method to assess vegetables producers: preferences for crop insurance. In 113th EAAE Seminar: a resilient European food chain in a challenging world.
- Meuwissen, M.P.M., de Mey, Y. and van Asseldonk, A.M. (2018). Prospects for agricultural insurance in Europe. Agricultural Finance Review, 78(2): 174-182.
- Michel-Kerjan, E. O. (2010). Catastrophe economics: the national flood insurance program. Journal of economic perspectives, 24(4): 165-86.
- Mishra, A. K., Nimon, R. W., and El-Osta, H. S. (2005). Is moral hazard good for the environment? Revenue insurance and chemical input use. Journal of environmental management, 74(1): 11-20.
- Möhring, N., Bozzola, M., Hirsch, S., and Finger, R. (2020a). Are pesticides risk decreasing? The relevance of pesticide indicator choice in empirical analysis. Agricultural Economics, 51(3): 429-444.

- Möhring, N., Dalhaus, T., Enjolras, G., and Finger, R. (2020b). Crop insurance and pesticide use in European agriculture. Agricultural Systems, 184, 102902.
- Möllmann, J., Michels, M., and Musshoff, O. (2019). German farmers' acceptance of subsidized insurance associated with reduced direct payments. Agricultural Finance Review, 79(3): 408-424.
- Moschini, G., and Hennessy, D.A. (2001). Uncertainty, risk aversion, and risk management for agricultural producers, in: Handbook of Agricultural Economics. Elsevier: 87–153. https://doi.org/10.1016/S1574-0072(01)10005-8
- Nielsen, T., Keil, A., and Zeller, M. (2013). Assessing farmers' risk preferences and their determinants in a marginal upland area of Vietnam: A comparison of multiple elicitation techniques. Agricultural Economics, 44(3): 255-273.
- Norton, M., van Sprundel, G. J., Turvey, C. G., and Meuwissen, M. P. (2016). Applying weather index insurance to agricultural pest and disease risks. International Journal of Pest Management, 62(3): 195-204.
- OECD (2022). Food Supply Chains and COVID-19: Impacts and Policy Lessons. OECD Policy Responses to Coronavirus (COVID-19). Policy Brief. 2 June 2020. Available at: https://www.oecd.org/coronavirus/policy-responses/food-supply-chains-and-covid-19-impacts-and-policy-lessons-71b57aea/ Accessed on: 09/11/2022.
- Paulson, N. D., and Babcock, B. A. (2010). Readdressing the fertilizer problem. Journal of Agricultural and Resource Economics, 35(3): 368-384.
- Pennings, J. M., and Leuthold, R. M. (2000). The role of farmers' behavioral attitudes and heterogeneity in futures contracts usage. American Journal of Agricultural Economics, 82(4): 908-919.
- Ramsey, F.P. (1931). Foundations of Mathematics and other Logical Essays. Routledge, London. https://doi. org/10.4324/9781315887814
- Reynaud, A., and Couture, S. (2012). Stability of risk preference measures: results from a field experiment on French farmers. Theory Decision 73: 203–221.
- Reynaud, A., Nguyen, M. H., and Aubert, C. (2018). Is there a demand for flood insurance in Vietnam? Results from a choice experiment. Environmental Economics and Policy Studies, 20: 1-25.
- Rippo, R., and Cerroni, S. (2022) Farmers' participation in the Income Stabilisation Tool: Evidence from the apple sector in Italy. Journal of Agricultural Economics, 74 (1): 273–294. https://doi.org/10.1111/1477-9552.12508
- Rogers, E. (1995). Diffusion of innovations (4th ed.). The Free Press, New York NY.

- Rommel, J., Hermann, D., Muller, M., and Musshoff, O. (2019). Contextual framing and monetary incentives in field experiments on risk preferences: Evidence from German farmers. Journal of Agricultural Economics, 70(2): 408–425
- Ross, D., and Williams, J. (2009). Basis risk from the cedant's perspective. In Pauline Barrieu and Luca Albertini, editors, The Handbook of Insurance-Linked Securities, chapter 5: 49–64.
- Ruto, E., and Garrod, G. (2009). Investigating farmers' preferences for the design of agri-environment schemes: a choice experiment approach. Journal of Environmental Planning and Management, 52(5): 631-647.
- Santeramo, F. G. (2019). I learn, you learn, we gain experience in crop insurance markets. Applied Economic Perspectives and Policy, 41(2): 284-304.
- Santeramo, F.G. (2018). Imperfect information and participation in insurance markets: evidence from Italy. Agricultural Finance Review, 78(2): 183–194.
- Santeramo, F.G., Goodwin, B.K., Adinolfi, F. and Capitanio, F. (2016). Farmer participation, entry and exit decisions in the Italian Crop Insurance Programme. Journal of Agricultural Economics, 67(3): 639–657.
- Santeramo, F. G., and Kang, M. (2022). Food Security Threats and Policy Responses in EU and Africa. Sustainable Horizons, 4, 100044.
- Santeramo, F. G., and Ramsey, A. (2017). Crop Insurance in the EU: Lessons and Caution from the US. Euro-Choices, 16(3): 34-39.
- Santeramo, F. G., Russo, I., and Lamonaca, E. (2022). Italian subsidised crop insurance: what the role of policy changes. Q Open. https://doi.org/10.1093/qopen/ qoac031
- Savage, L.J. (1954). The foundations of statistics, John Wiley & Sons, Inc., Naval Research Logistics Quarterly, 1: 236-236. https://doi.org/10.1002/nav.3800010316
- Serra, T., Zilberman, D., and Gil, J. M. (2008). Farms' technical inefficiencies in the presence of government programs. Australian Journal of Agricultural and Resource Economics, 52(1): 57–76.
- Serra, T., Zilberman, D., Goodwin, B. K., and Featherstone, A. (2006). Effects of decoupling on the mean and variability of output. European Review of Agricultural Economics, 33: 269–288.
- Severini, S., Di Tommaso, G. and Finger, R. (2019). Effects of the Income Stabilization Tool on farm income level, variability and concentration in Italian agriculture. Agricultural and Food Economics, 7(1): 1-23.
- Severini, S., Tantari, A., and Di Tommaso, G. (2017). Effect of agricultural policy on income and revenue

risks in Italian farms: implications for the use of risk management policies. Agricultural Finance Review, 77(2): 295-311

- Sorvali, J., Kaseva, J., and Peltonen-Sainio, P. (2021). Farmer views on climate change—a longitudinal study of threats, opportunities and action. Climatic Change, 164: 1-19.
- Spiegel A., Soriano B., de Mey Y., Slijper T., Urquhart J., Bardají I., Vigani M., Severini S., and Meuwissen M. (2020). Risk Management and its Role in Enhancing Perceived Resilience Capacities of Farms and Farming Systems in Europe. EuroChoices, 19(2): 45-52. https://doi.org/10.1111/1746-692X.12284.
- SriRamaratnam, S., D. A. Bessler, M. E. Rister, J. E. Matocha, and J. Novak. (1987). Fertilization Under Uncertainty: An Analysis Based on Producer Yield Expectations. American Journal of Agricultural Economics, 69(2): 349–357.
- Stigler, M., and Lobell, D. (2022). Optimal index insurance and basis risk decomposition: an application to Kenya. American Journal of Agricultural Economics, Early View. https://doi.org/10.1111/ajae.12375
- Tang, Y., Cai, H., and Liu, R. (2022). Will marketing strategies affect farmers' preferences and willingness to pay for catastrophe insurance? Evidence from a choice experiment in China. Environment, Development and Sustainability, 24(1): 1376-1389.
- Tornatzky L., and Fleischer M. (1990). The Process of Technology Innovation. Lexington, MA, Lexington Books.
- Trestini, S., Giampietri, E., and Boatto V. (2017). Toward the implementation of the Income Stabilization Tool: an analysis of factors affecting the probability of farm income losses in Italy, New Medit, 16(4): 24-30.
- Tversky, A. and D. Kahneman (1992). Advances in prospect theory: Cumulative representation of uncertainty. Journal of Risk and Uncertainty, 5(4): 297–323.
- Ullman, J. B., and Bentler, P. M. (2003). Structural equation modeling. Handbook of psychology: 607-634.
- van Winsen, F., de Mey, Y., Lauwers, L., Van Passel, S., Vancauteren, M., and Wauters, E. (2016). Determinants of risk behaviour: effects of perceived risks and risk attitude on farmer's adoption of risk management strategies. Journal of Risk Research, 19(1): 56-78.
- Vedenov, D. V., Epperson, J. E., and Barnett, B. J. (2006). Designing catastrophe bonds to securitize systemic risks in agriculture: the case of Georgia cotton. Journal of Agricultural and Resource Economics, 31(2): 318-338.
- Venkatesh, V., Morris, M. G., Davis, G. B., and Davis, F. D. (2003). User acceptance of information technology: Toward a unified view. MIS quarterly, 425-478.

- von Neumann, J., and Morgenstern, O. (1947). Theory of games and economic behavior. Princeton University Press, Princeton, NJ.
- Vroege, W., Dalhaus, T., and Finger, R. (2019). 'Index insurances for grasslands – A review for Europe and North-America'. Agricultural Systems 168: 101-111
- Ward, P. S., and Makhija, S. (2018). New modalities for managing drought risk in rainfed agriculture: Evidence from a discrete choice experiment in Odisha, India. World Development, 107: 163-175.
- Was, A., and Kobus, P. (2018). Factors differentiating the level of crop insurance at Polish farms. Agricultural Finance Review, 78(2): 209–222.
- Zinnanti, C., Coletta, A., Torrigiani, M., and Severini, S. (2022). Simulating risk management strategies for specialized farming systems: the potential impact of the EU income stabilization tool. Agricultural Finance Review, 83(2): 264-285. https://doi. org/10.1108/AFR-11-2021-0146